Author's response to reviews

Title: Ventilatory abnormalities in patients with cystic fibrosis undergoing submaximal exercise test six minutes

Authors:

Paloma p.l.f.P Lopes Francisco Parazzi (paloma.francisco@hotmail.com)
Fernando f.a.l.M Augusto de Lima Marson (fernandolimamarson@hotmail.com)
Maria Angela m.a.g.o.R Gonçalves de Oliveira Ribeiro (ribeiromago@gmail.com)
Celíce c.c.b.A Cruz Bresciani de Almeida (ccb.almeida@gmail.com)
Luis Claudio l.c.M Martins (lclaudio@fcm.unicamp.br)
Ilma i.a.P Aparecida Pascoal (Ilma@mpc.com.br)
Adyléia a.a.d.c.T Aparecida Dalbo Contrera Toro (dalbotoro@terra.com.br)
Camila c.i.s.S Isabel Santos Shivinski (cacaiss@yahoo.com.br)
José Dirceu j.d.R Ribeiro (jdirceuribeiro@gmail.com)

Version: 2
Date: 1 June 2014

Author's response to reviews:

1- Thanks were detailed.

2- Lines and numbers of pages were included

3- The name of the Ethics Committee that approved the study was included in the review.

4- The conclusion was included after the discussion